Author's response to reviews

Title: Unusual Causes of Parkinsonism: a case series

Authors:

Solomon O Ugoya Dr (docsouls@yahoo.com)
Emmanuel I Agaba Dr (eiagaba@yahoo.com)
Comfort A Daniyam Dr (comfort_13@yahoo.com)

Version: 2 Date: 8 April 2010

Author's response to reviews:

The Editor in Chief:
We have made changes using the prototype you included in your letter
1. The abstract was completely changed in accordance with the Journal style.
2. Mechanism of action of drugs was included in the introduction and referenced.
3. Conclusion was also changed.
4. Addition of statement on consent and conflict of interest was added.
5. Authors contribution was also included.
6. References were added and change appropriately.